Author's response to reviews

Title: Learning to improve medical decision making from imbalanced data without a priori cost

Authors:

Xiang Wan (xwan@comp.hkbu.edu.hk)
Jiming Liu (jiming@comp.hkbu.edu.hk)
William K. Cheung (william@comp.hkbu.edu.hk)
Tiejun Tong (tongt@hkbu.edu.hk)

Version: 3
Date: 5 November 2014

Author's response to reviews: see over
Response to Editor’s comments:

1. *Please include all authors’ contact and institutions in the first page of the manuscript.*

**Answer:** We include all authors’ contact and institutions at the beginning. But the latex template of BioMed Central only keeps the first author’s information in the first page and put all the others at the end.

2. *‘Experiment’ section should be removed and fused with ‘Methods’.*

**Answer:** We have done it in the revised manuscript.

3. *‘Results’ section should be introduced.*

**Answer:** We have introduced this section in the revised manuscript.

4. *Please clearly state in the text that the data sets use were publicly available.*

**Answer:** Thanks for the comments. We have clearly made this statement in the ‘Data Preparation’ subsection in our revised manuscript.